%Title: Determination of Spin and CP of the Higgs boson from WBF
%Author: C.-P. Buszello, P. Marquard
%hep-ph/0603209
\documentclass[notoc,paper,12pt]{JHEP3}
   
\usepackage{amsmath}
\usepackage{graphicx}
\author{C.-P.~Buszello\\ Cavendish Laboratory, University of Cambridge, \\
  E-mail: \email{buszello@hep.phy.cam.ac.uk}}

\author{P.~Marquard\\ 
  Institut f\"ur Theoretische Teilchenphysik, Universit\"at Karlsruhe, \\
  E-mail: \email{Marquard@particle.uni-karlsruhe.de}}

\preprint{TTP06-10\\
SFB/CPP-06-11\\
CAVENDISH-HEP-06-10\\
hep-ph/0603209\\}

\keywords{Higgs Physics, Hadronic Colliders, Standard Model, Phenomenological Models}
\title{Determination of Spin and CP of the Higgs Boson from WBF}

\abstract{We explore the possibilities to determine the spin/CP
  properties of the Higgs boson at the LHC. To cover the mass region
  below the ZZ threshold we make use of the properties of the production in Weak Boson Fusion (WBF)
  and the decay chain H $\rightarrow$ W$^+$W$^-$ $\rightarrow$ $\ell^+\nu \ell^-\nu$. 
  In particular, we study the angular correlations of the 
  forward jets and the distribution of the invariant mass of the lepton pair for
  different hypothetical Higgs like particles.}
\begin{document}
\begin{thebibliography}{99}
%\cite{Eidelman:2004wy}
\bibitem{Eidelman:2004wy}
  S.~Eidelman {\it et al.}  [Particle Data Group],
  %``Review of particle physics,''
  Phys.\ Lett.\ B {\bf 592} (2004) 1.
  %%CITATION = PHLTA,B592,1;%%

%\cite{Dittmar:1996ss}
\bibitem{Dittmar:1996ss}
  M.~Dittmar and H.~K.~Dreiner,
  %``How to find a Higgs boson with a mass between 155-GeV to 180-GeV at the
  %LHC,''
  Phys.\ Rev.\ D {\bf 55} (1997) 167
  [arXiv:hep-ph/9608317].
  %%CITATION = HEP-PH 9608317;%%

%\cite{Choi:2002jk}
\bibitem{Choi:2002jk}
S.~Y.~Choi, D.~J.~Miller, M.~M.~Muhlleitner and P.~M.~Zerwas,
%``Identifying the Higgs spin and parity in decays to Z pairs,''
Phys.\ Lett.\ B {\bf 553} (2003) 61
[arXiv:hep-ph/0210077].
%%CITATION = HEP-PH 0210077;%%

%\cite{Rainwater:1999sd}
\bibitem{Rainwater:1999sd}
  D.~L.~Rainwater and D.~Zeppenfeld,
  %``Observing $H \to W^{(*)}W^{(*)} \to e^\pm \mu^\mp /\!\!\!{p}_T$ in weak
  %boson fusion with dual forward jet tagging at the CERN LHC,''
  Phys.\ Rev.\ D {\bf 60}, 113004 (1999)
  [Erratum-ibid.\ D {\bf 61}, 099901 (2000)]
  [arXiv:hep-ph/9906218].
  %%CITATION = HEP-PH 9906218;%%

%\cite{Asai:2004ws}
\bibitem{Asai:2004ws}
  S.~Asai {\it et al.},
  %``Prospects for the search for a standard model Higgs boson in ATLAS  using
  %vector boson fusion,''
  Eur.\ Phys.\ J.\ C {\bf 32S2} (2004) 19
  [arXiv:hep-ph/0402254].
  %%CITATION = HEP-PH 0402254;%%

\bibitem{cms-higgs} S.~Abdullin et al., 
%{\em Summary of the CMS potential for the Higgs boson discovery}, 
 CMS NOTE 2003/033.

%\cite{Buszello:2004be}
\bibitem{Buszello:2004be}
C.~P.~Buszello, P.~Marquard and J.~J.~van der Bij,
%``On the determination of the structure of the scalar Higgs boson's couplings
%to vectorbosons,''
arXiv:hep-ph/0406181.
%%CITATION = HEP-PH 0406181;%%

%\cite{Buszello:2002uu}
\bibitem{Buszello:2002uu}
C.~P.~Buszello, I.~Fleck, P.~Marquard and J.~J.~van der Bij,
%``Prospective analysis of spin- and CP-sensitive variables in H $\to$ Z Z $\to$
%l(1)+ l(1)- l(2)+ l(2)- at the LHC,''
Eur.\ Phys.\ J.\ C {\bf 32} (2004) 209
[arXiv:hep-ph/0212396].
%%CITATION = HEP-PH 0212396;%%

%\cite{Figy:2003nv}
\bibitem{Figy:2003nv}
T.~Figy, C.~Oleari and D.~Zeppenfeld,
%``Next-to-leading order jet distributions for Higgs boson production via
%weak-boson fusion,''
Phys.\ Rev.\ D {\bf 68} (2003) 073005
[arXiv:hep-ph/0306109].
%%CITATION = HEP-PH 0306109;%%

%\cite{DelDuca:2003ba}
\bibitem{DelDuca:2003ba}
V.~Del Duca, W.~Kilgore, C.~Oleari, C.~R.~Schmidt and D.~Zeppenfeld,
%``Kinematical limits on Higgs boson production via gluon fusion in association
%with jets,''
Phys.\ Rev.\ D {\bf 67} (2003) 073003
[arXiv:hep-ph/0301013].
%%CITATION = HEP-PH 0301013;%%

%\cite{Maltoni:2002qb}
\bibitem{Maltoni:2002qb}
  F.~Maltoni and T.~Stelzer,
  %``MadEvent: Automatic event generation with MadGraph,''
  JHEP {\bf 0302} (2003) 027
  [arXiv:hep-ph/0208156].
  %%CITATION = HEP-PH 0208156;%%

%\cite{Sjostrand:2003wg}
\bibitem{Sjostrand:2003wg}
  T.~Sjostrand, L.~Lonnblad, S.~Mrenna and P.~Skands,
  %``PYTHIA 6.3: Physics and manual,''
  arXiv:hep-ph/0308153.
  %%CITATION = HEP-PH 0308153;%%

\end{thebibliography}


\input{introduction}

\input{chapter3}

\section{Conclusions}
We have implemented a matrix element generator capable of simulating the
generation of Higgs bosons via WBF and its decay to W pairs with SM and
non SM spin and CP quantum numbers (namely $0^{+}, 0^{-}, 1^+, 1^-$). We
have demonstrated that the different couplings lead to distinct
distributions of the opening angle of the tag jets and the invariant
mass of the leptons from the leptonic decay of the W pairs. The latter
distributions are very sensitive to the lepton cuts used to isolate the
signal.  This leads to the necessity of optimizing these cuts carefully
when performing the measurement using real data, depending on the
background level observed in the experiment. The tag jet distributions
appear much more robust against the cuts. They provide very promising
prospects to confirm the spin/CP state of an SM Higgs with a mass
between 130 and 180 GeV using an integrated luminosity of 30fb$^{-1}$.



\end{document}
